\begin{thebibliography}{10}

\bibitem{endres03}
Rombach A.~Endres, H.D.
\newblock {\em A Handbook of Software and Systems Engineering: Empirical
  Observations, Laws and Theories}.
\newblock Addison Wesley, 2003.

\bibitem{aamod94}
A.~Aamodt and E.~Plaza.
\newblock Case-based reasoning: Foundational issues, methodological variations,
  and system approaches.
\newblock {\em Artificial Intellegence Communications}, 7:39--59, 1994.

\bibitem{andrews07}
J.H. Andrews, F.C.H. Li, and T.~Menzies.
\newblock Nighthawk: A two-level genetic-random unit test data generator.
\newblock In {\em IEEE ASE'07}, 2007.
\newblock Available from \url{http://menzies.us/pdf/07ase-nighthawk.pdf}.

\bibitem{azzeh08}
Mohammad Azzeh, Daniel Neagu, and Peter Cowling.
\newblock Improving analogy software effort estimation using fuzzy feature
  subset selection algorithm.
\newblock In {\em PROMISE '08: Proceedings of the 4th international workshop on
  Predictor models in software engineering}, pages 71--78, 2008.

\bibitem{baker07}
Dan Baker.
\newblock A hybrid approach to expert and model-based effort estimation.
\newblock Master's thesis, Lane Department of Computer Science and Electrical
  Engineering, West Virginia University, 2007.
\newblock Available from
  \url{https://eidr.wvu.edu/etd/documentdata.eTD?documentid=5443}.

\bibitem{bartlett32}
F.C. Bartlett.
\newblock {\em Remembering: A study in experimental and social psychology}.
\newblock The Cambridge University Press, 1932.

\bibitem{boehm81}
B.~Boehm.
\newblock {\em Software Engineering Economics}.
\newblock Prentice Hall, 1981.

\bibitem{boehm00a}
B.~Boehm.
\newblock Safe and simple software cost analysis.
\newblock {\em IEEE Software}, pages 14--17, September/October 2000.
\newblock Available from
  \url{http://www.computer.org/certification/beta/Boehm_Safe.pdf}.

\bibitem{Boehm96softwarecost}
Barry Boehm, Barry Boehm, and Hoh In.
\newblock Software cost option strategy tool (s-cost).
\newblock In {\em In Conflict Analysis and Negotiation Aids for Cost-Quality
  Requirements Annual International Computer Software and Applications
  Conference), IEEE Comp}, pages 15--20. Society Press, 1996.

\bibitem{boehm00b}
Barry Boehm, Ellis Horowitz, Ray Madachy, Donald Reifer, Bradford~K. Clark,
  Bert Steece, A.~Winsor Brown, Sunita Chulani, and Chris Abts.
\newblock {\em Software Cost Estimation with Cocomo II}.
\newblock Prentice Hall, 2000.

\bibitem{brady10b}
Adam Brady and Tim Menzies.
\newblock Case-based reasoning vs parametric models for software quality
  optimization.
\newblock In {\em PROMISE'10}, 2010.
\newblock Available from \url{http://menzies.us/pdf/10cbr.pdf}.

\bibitem{brady10a}
Adam Brady, Tim Menzies, Oussama El-Rawas, Ekrem Kocaguneli, and Jacky Keung.
\newblock Case-based reasoning for reducing software development effort.
\newblock {\em Journal of Software Engineering and Applications}, 3, 2010.
\newblock Available from \url{http://menzies.us/pdf/10w0.pdf}.

\bibitem{brooks75}
F.~P. Brooks.
\newblock {\em The Mythical Man-Month, Anniversary edition}.
\newblock Addison-Wesley, 1975.

\bibitem{chulani99}
S.~Chulani, B.~Boehm, and B.~Steece.
\newblock Bayesian analysis of empirical software engineering cost models.
\newblock {\em IEEE Transaction on Software Engineerining}, 25(4), July/August
  1999.

\bibitem{craw94}
S.~Craw, D.~Sleeman, R.~Boswell, and L.~Carbonara.
\newblock Is knowledge refinement different from theory revision?
\newblock In S.~Wrobel, editor, {\em Proceedings of the MLNet Familiarization
  Workshop on Theory Revision and Restructuring in Machine Learning (ECML-94)},
  pages 32--34, 1994.

\bibitem{Dil84}
W.~Dillon and M.~Goldstein.
\newblock {\em Multivariate Analysis: Methods and Applications}.
\newblock Wiley-Interscience, 1984.

\bibitem{me10f}
Oussama El-Rawas and Tim Menzies.
\newblock A second look at faster, better, cheaper.
\newblock {\em Innovations in Systems and Software Engineering}, 2011.

\bibitem{fenton99}
N.~E. Fenton and M.~Neil.
\newblock A critique of software defect prediction models.
\newblock {\em IEEE Transactions on Software Engineering}, 25(5):675--689,
  1999.
\newblock Available from
  \url{http://citeseer.nj.nec.com/fenton99critique.html}.

\bibitem{fenton07}
Norman~E. Fenton, Martin Neil, and Jose~Galan Caballero.
\newblock Using ranked nodes to model qualitative judgments in bayesian
  networks.
\newblock {\em IEEE Trans. on Knowl. and Data Eng.}, 19(10):1420--1432, 2007.

\bibitem{gof95}
E.~Gamma, R.~Helm, R.~Johnson, and J.~Vlissides.
\newblock {\em Design Patterns: Elements of Reusable Object-Oriented Software}.
\newblock Addison-Wesley, 1995.

\bibitem{me09i}
P.~Green, T.~Menzies, S.~Williams, and O.~El-waras.
\newblock Understanding the value of software engineering technologies.
\newblock In {\em IEEE ASE'09}, 2009.
\newblock Available from \url{http://menzies.us/pdf/09value.pdf}.

\bibitem{hall03}
M.A. Hall and G.~Holmes.
\newblock Benchmarking attribute selection techniques for discrete class data
  mining.
\newblock {\em IEEE Transactions On Knowledge And Data Engineering},
  15(6):1437-- 1447, 2003.
\newblock Available from
  \url{http://www.cs.waikato.ac.nz/~mhall/HallHolmesTKDE.pdf}.

\bibitem{harman04}
Mark Harman and Joachim Wegener.
\newblock Getting results from search-based approaches to software engineering.
\newblock In {\em ICSE '04: Proceedings of the 26th International Conference on
  Software Engineering}, pages 728--729, Washington, DC, USA, 2004. IEEE
  Computer Society.

\bibitem{holzmann97}
G.J. Holzmann.
\newblock The model checker {SPIN}.
\newblock {\em IEEE Transactions on Software Engineering}, 23(5):279--295, May
  1997.

\bibitem{jalali08}
O.~Jalali, T.~Menzies, and M.~Feather.
\newblock Optimizing requirements decisions with keys.
\newblock In {\em Proceedings of the PROMISE 2008 Workshop (ICSE)}, 2008.
\newblock Available from \url{http://menzies.us/pdf/08keys.pdf}.

\bibitem{jorgensen05}
M.~Jorgensen and M.~Shepperd.
\newblock A systematic review of software development cost estimation studies,
  January 2007.
\newblock Available from
  \url{http://www.simula.no/departments/engineering/publications/Jorgensen.200%
5.12}.

\bibitem{kemerer87}
C.F. Kemerer.
\newblock An empirical validation of software cost estimation models.
\newblock {\em Communications of the ACM}, 30(5):416--429, May 1987.

\bibitem{journals/ese/KhoshgoftaarS03}
Taghi~M. Khoshgoftaar and Naeem Seliya.
\newblock Fault prediction modeling for software quality estimation: Comparing
  commonly used techniques.
\newblock {\em Empirical Software Engineering}, 8(3):255--283, 2003.

\bibitem{kirkpatrick83}
S.~Kirkpatrick, C.~D. Gelatt, and M.~P. Vecchi.
\newblock Optimization by simulated annealing.
\newblock {\em Science, Number 4598, 13 May 1983}, 220, 4598:671--680, 1983.

\bibitem{Kirsopp2002}
C.~Kirsopp and M.~Shepperd.
\newblock {Making inferences with small numbers of training sets}.
\newblock {\em IEEE Proc.}, 149, 2002.

\bibitem{me10c}
Ekrem Kocaguneli, Gregory Gay, Tim Menzies, Ye~Yang, and Jacky~W. Keung.
\newblock When to use data from other projects for effort estimation.
\newblock In {\em IEEE ASE'10}, 2010.
\newblock Available from \url{http://menzies.us/pdf/10other.pdf}.

\bibitem{kol93}
J.~Kolodner.
\newblock {\em Case-Based Reasoning}.
\newblock Morgan Kaufmann, 1993.

\bibitem{kolodner83}
Janet Kolodner.
\newblock Reconstructive memory: A computer model.
\newblock {\em Cognitive Science}, 7(4):281--328, 1983.

\bibitem{leake05}
David Leake and David Mcsherry.
\newblock Intro. to the special issue on explanation in case-based reasoning.
\newblock {\em AI Review}, 24:103--108, 2005.

\bibitem{leake96}
David~B. Leake.
\newblock {\em Case-Based Reasoning: Experiences, Lessons and Future
  Directions}.
\newblock MIT Press, Cambridge, MA, USA, 1996.

\bibitem{Li2009}
Y~Li, M~Xie, and T~Goh.
\newblock {A study of project selection and feature weighting for analogy based
  software cost estimation}.
\newblock {\em Journal of Systems and Software}, 82:241--252, 2009.

\bibitem{Lipowezky1998}
U.~Lipowezky.
\newblock {Selection of the optimal prototype subset for 1-NN classification}.
\newblock {\em Pattern Recognition Letters}, 19:907918, 1998.

\bibitem{loftus03}
E.F. Loftus.
\newblock Our changeable memories: legal and practical implications.
\newblock {\em Nature Rev. Neurosci.}, pages 231--234, 2003.

\bibitem{lowry10}
Michael~R. Lowry.
\newblock Towards predictive models of technology impact on software design
  productivity.
\newblock 2010.

\bibitem{Mendes2003}
Emilia Mendes, Ian~D. Watson, Chris Triggs, Nile Mosley, and Steve Counsell.
\newblock A comparative study of cost estimation models for web hypermedia
  applications.
\newblock {\em Empirical Software Engineering}, 8(2):163--196, 2003.

\bibitem{me09f}
T.~Menzies, O.~El-Rawas, J.~Hihn, and B.~Boehm.
\newblock Can we build software faster and better and cheaper?
\newblock In {\em PROMISE'09}, 2009.
\newblock Available from \url{http://menzies.us/pdf/09bfc.pdf}.

\bibitem{me07g}
T.~Menzies, O.~Elrawas, D.~Baker, J.~Hihn, and K.~Lum.
\newblock On the value of stochastic abduction (if you fix everything, you lose
  fixes for everything else).
\newblock In {\em International Workshop on Living with Uncertainty (an ASE'07
  co-located event)}, 2007.
\newblock Available from \url{http://menzies.us/pdf/07fix.pdf}.

\bibitem{me08c}
T.~Menzies, O.~Elrawas, B.~Barry, R.~Madachy, J.~Hihn, D.~Baker, and K.~Lum.
\newblock Accurate estimates without calibration.
\newblock In {\em International Conference on Software Process}, 2008.
\newblock Available from \url{http://menzies.us/pdf/08icsp.pdf}.

\bibitem{me07f}
T.~Menzies, O.~Elrawas, J.~Hihn, M.~Feathear, B.~Boehm, and R.~Madachy.
\newblock The business case for automated software engineerng.
\newblock In {\em ASE '07: Proceedings of the twenty-second IEEE/ACM
  international conference on Automated software engineering}, pages 303--312,
  New York, NY, USA, 2007. ACM.
\newblock Available from \url{http://menzies.us/pdf/07casease-v0.pdf}.

\bibitem{me01c}
T.~Menzies and J.D. Kiper.
\newblock How to argue less, 2001.
\newblock Available from \url{http://menzies.us/pdf/01jane.pdf}.

\bibitem{me09a}
T.~Menzies, S.~Williams, O.~El-rawas, B.~Boehm, and J.~Hihn.
\newblock How to avoid drastic software process change (using stochastic
  statbility).
\newblock In {\em ICSE'09}, 2009.
\newblock Available from \url{http://menzies.us/pdf/08drastic.pdf}.

\bibitem{me09e}
T.~Menzies, S.~Williams, O.~Elrawas, D.~Baker, B.~Boehm, J.~Hihn, K.~Lum, and
  R.~Madachy.
\newblock Accurate estimates without local data?
\newblock {\em Software Process Improvement and Practice}, 14:213--225, July
  2009.
\newblock Available from \url{http://menzies.us/pdf/09nodata.pdf}.

\bibitem{me05a}
Tim Menzies, Zhihao Chen, Dan Port, and Jairus Hihn.
\newblock Simple software cost estimation: Safe or unsafe?
\newblock In {\em Proceedings, PROMISE workshop, ICSE 2005}, 2005.
\newblock Available from \url{http://menzies.us/pdf/05safewhen.pdf}.

\bibitem{me02f}
Tim Menzies, David Raffo, Siri on~Setamanit, Ying Hu, and Sina Tootoonian.
\newblock Model-based tests of truisms.
\newblock In {\em Proceedings of IEEE ASE 2002}, 2002.
\newblock Available from \url{http://menzies.us/pdf/02truisms.pdf}.

\bibitem{me10b}
Tim Menzies and Forrest Shull.
\newblock The quest for convincing evidence.
\newblock In A.~Oram and G.Wilson, editors, {\em Making Software: What Really
  Works and We We Believe it}. O'Reilly Books, 2010.

\bibitem{me00f}
T.J. Menzies.
\newblock The complexity of trmcs-like spiral specification.
\newblock In {\em Proceedings of 10th International Workshop on Software
  Specification and Design (IWSSD-10)}, 2000.
\newblock Available from \url{http://menzies.us/pdf/00iwssd.pdf}.

\bibitem{me10a}
T.J. Menzies, Z.~Milton, B.~Turhan, B.~Cukic, Y.~Jiang, and A.~Bener.
\newblock Defect prediction from static code features: Current results,
  limitations, new approaches.
\newblock {\em Automated Software Engineering}, (4), December 2010.
\newblock Available from \url{http://menzies.us/pdf/10which.pdf}.

\bibitem{miller02}
A.~Miller.
\newblock {\em Subset Selection in Regression (second edition)}.
\newblock Chapman \& Hall, 2002.

\bibitem{miyazaki94}
Y.~Miyazaki, M.~Terakado, K.~Ozaki, and H.~Nozaki.
\newblock Robust regression for developing software estimation models.
\newblock {\em J. Syst. Softw.}, 27(1):3--16, 1994.

\bibitem{mozina04}
Martin Mo\v{z}ina, Janez Dem\v{s}ar, Michael Kattan, and Bla\v{z} Zupan.
\newblock Nomograms for visualization of naive bayesian classifier.
\newblock In {\em PKDD '04: Proceedings of the 8th European Conference on
  Principles and Practice of Knowledge Discovery in Databases}, pages 337--348,
  New York, NY, USA, 2004. Springer-Verlag New York, Inc.

\bibitem{ruhe09}
An~Ngo-The and G.~Ruhe.
\newblock Optimized resource allocation for software release planning.
\newblock {\em Software Engineering, IEEE Transactions on}, 35(1):109--123,
  Jan.-Feb. 2009.

\bibitem{orrego09}
A.~Orrego, T.~Menzies, and O.~El-Rawas.
\newblock On the relative merits of software reuse.
\newblock In {\em International Conference on Software Process}, 2009.
\newblock Available from \url{http://menzies.us/pdf/09reuse.pdf}.

\bibitem{ostrand04}
Thomas~J. Ostrand, Elaine~J. Weyuker, and Robert~M. Bell.
\newblock Where the bugs are.
\newblock In {\em ISSTA '04: Proceedings of the 2004 ACM SIGSOFT international
  symposium on Software testing and analysis}, pages 86--96, New York, NY, USA,
  2004. ACM.

\bibitem{pendharkar05}
Parag~C. Pendharkar, Girish~H. Subramanian, and James~A. Rodger.
\newblock A probabilistic model for predicting software development effort.
\newblock {\em IEEE Trans. Softw. Eng.}, 31(7):615--624, 2005.

\bibitem{pfahl05}
Dietmar Pfahl and Ioana Rus.
\newblock Special issue on prosim 2004.
\newblock {\em Software Process: Improvement and Practice}, 10(3):251--253,
  July/September.

\bibitem{port08}
D.~Port, A.~Olkov, and T.~Menzies.
\newblock Using simulation to investigate requirements prioritization
  strategies.
\newblock In {\em IEEE ASE'08}, 2008.
\newblock Available from \url{http://menzies.us/pdf/08simrequire.pdf}.

\bibitem{norvig03}
Stuart~J. Russell, Peter Norvig, John~F. Candy, Jitendra~M. Malik, and
  Douglas~D. Edwards.
\newblock {\em Artificial intelligence: a modern approach}.
\newblock Prentice-Hall, Inc., Upper Saddle River, NJ, USA, 2003.

\bibitem{schank83}
Roger~C. Schank.
\newblock {\em Dynamic Memory: A Theory of Reminding and Learning in Computers
  and People}.
\newblock Cambridge University Press, New York, NY, USA, 1983.

\bibitem{shank77}
Roger~C. Schank and Robert~P. Abelson.
\newblock {\em Scripts, plans, goals and understanding: an inquiry into human
  knowledge structures}.
\newblock Erlbaum, 1977.

\bibitem{schulz10}
Thomas Schulz, Lukasz Radlinski, Thomas Gorges, and Wolfgang Rosenstiel.
\newblock Defect cost flow model: a bayesian network for predicting defect
  correction effort.
\newblock In {\em PROMISE '10}, pages 16:1--16:11, 2010.

\bibitem{selman96}
Bart Selman, Henry~A. Kautz, and Bram Cohen.
\newblock Local search strategies for satisfiability testing.
\newblock In Michael Trick and David~Stifler Johnson, editors, {\em Proceedings
  of the Second {DIMACS} Challange on Cliques, Coloring, and Satisfiability},
  Providence RI, 1993.

\bibitem{shepperd07}
M.~Shepperd.
\newblock Software project economics: A roadmap.
\newblock In {\em International Conference on Software Engineering 2007: Future
  of Software Engineering}, 2007.

\bibitem{shepperd97}
M.~Shepperd and C.~Schofield.
\newblock Estimating software project effort using analogies.
\newblock {\em IEEE Transactions on Software Engineering}, 23(12), November
  1997.
\newblock Available from \url{http://www.utdallas.edu/~rbanker/SE_XII.pdf}.

\bibitem{She02}
M.~J. Shepperd.
\newblock Case-based reasoning and software engineering.
\newblock Technical Report TR02-08, Bournemouth University, UK, 2002.

\bibitem{zimmermann09}
H.~Gall E.~Giger T.~Zimmermann, N.~Nagappan and B.~Murphy.
\newblock Cross-project defect prediction.
\newblock In {\em ESEC/FSE'09}, August 2009.

\bibitem{tosun10}
A.~Tosun, A.~Bener, and R.~Kale.
\newblock Ai-based software defect predictors: Applications and benefits in a
  case study.
\newblock In {\em Twenty-Second IAAI Conference on Artificial Intelligence},
  2010.

\bibitem{Walkerden1999}
Fiona Walkerden and Ross Jeffery.
\newblock An empirical study of analogy-based software effort estimation.
\newblock {\em Empirical Softw. Engg.}, 4(2):135--158, 1999.

\bibitem{watson98}
Ian Watson.
\newblock {\em Applying case-based reasoning: techniques for enterprise
  systems}.
\newblock Morgan Kaufmann Publishers Inc., San Francisco, CA, USA, 1998.

\bibitem{witten05}
Ian~H. Witten and Eibe Frank.
\newblock {\em Data mining. 2nd edition}.
\newblock Morgan Kaufmann, Los Altos, US, 2005.

\bibitem{wolpert94}
David~H. Wolpert and R.~Waters.
\newblock The relationship between pac, the statistical physics framework, the
  bayesian framework, and the vc framework.
\newblock In {\em Proc. SFI/CNL Workshop Formal Approaches to Supervised
  Learning}, pages 117--214. Addison-Wesley, 1994.

\bibitem{zhang07a}
Y.~Zhang, M.~Harman, and S.A. Mansouri.
\newblock The multi-objective next release problem.
\newblock In {\em In ACM Genetic and Evolutionary Computation Conference (GECCO
  2007)}, page~11, 2007.

\end{thebibliography}

\section{Datasets and Project Descriptions}

\begin{figure}[!h]
\centering
\begin{tabular}{l|rrl|l}  
 \textbf{Dataset} & \textbf{Cols} & \textbf{Rows} & \textbf{Notes}&\textbf{Measures }\\\hline
Kemerer	& 7 & 15 & Large business applications	& effort \\ 
Telecom	& 3 & 18 & U.K. telecom enhancements	& effort\\ 
Finnish	& 8 & 38 & Finnish IS projects		& effort \\
Miyazaki& 8 & 48 & Japanese COBOL projects	& effort\\ 
COC81ii	&26 & 63 & NASA projects 		& effort, time, defects\\
NASA93ii	&26 & 93 & NASA projects 		& effort, time, defects \\
China	&17 &499 & Chinese software projects	& effort \\\cline{3-3}
          \multicolumn{2}{c}{~}  & Total: 774\\
\end{tabular}
\caption{Seven data sets from \protect\url{promisedata.org/?cat=14}: {\em effort} is total staff person-months; {\em  time} is calendar time (start to stop);
{\em defects} represents the number of delivered defects.
}\label{fig:data}
\end{figure}

Because {\W}2 makes no underlying model assumptions, we aren't limited to the COCOMO II ontology for our performance evaluations. However, in order to compare {\W}2's case-based methodology with SEESAW's model-based methodology, these comparisons must be made within the same ontology. For the following experiments, any comparison with SEESAW will use the NASA93 and COC81 datasets.

Both NASA93 and COC81 initially contained only effort data in the original COCOMO format. However, using Boehm's COQUALMO defect prediction we have converted this data into COCOMO II format, which contains estimated total defect numbers as well as estimated completion time in months. COCOMO II also breaks down the project attributes into both scale factors and effort multipliers, outlined earlier in ~\fig{cocattrs}.

The other datasets use for these experiments is detailed in ~\fig{data}. For non-COCOMO data, the reduction goal is the same: software effort in person-months. These datasets include the following:
\bi
\item
Enhancements to a U.K. telecommunications product;
\item
Projects collected by Miyazaki et al~\cite{miyazaki94};
\item
Finnish Information Systems projects;
\item
A large dataset of Chinese software projects;
\item
Large COBOL projects, collected by Kemerer~\cite{kemerer87}.
\ei
The format of this data is highly varied and includes number of basic logical transactions, query count and number of distinct business units serviced. 

Project descriptions are also needed to define the space of controllable options. For the non-COCOMO data sets, we did not have access to specific case studies like \fig{cases}. Hence, these results are based on {\em contexts} developed as follows:

\bi
\item One project contained all possible project values as $controllables$
\item The remaining two projects featured randomly chosen ranges with ranges half of the possible values.
\ei
Data and project descriptions for the NASA case studies defined in \fig{cases} are shown in ~\tion{dataset}. A project description file for a non-COCOMO dataset is shown in \fig{china-p2}.

\begin{figure}
\begin{verbatim}
@project china-Proj2
@attribute ?AFP vl lo md
@attribute ?Input md hi vh
@attribute ?Output vl lo md 
@attribute ?Enquiry md hi vh
@attribute ?File vl lo md
@attribute ?Interface hi vh
@attribute ?Added vl lo md
@attribute ?Changed md hi vh
@attribute ?Deleted vl lo md
@attribute ?PDR_AFP md hi vh
@attribute ?PDR_UFP md hi vh
@attribute ?NPDR_AFP vl lo md
@attribute ?NPDU_UFP vl lo md
@attribute ?Resource md hi
@attribute ?Dev.Type vl
@attribute ?Duration vl lo md hi
\end{verbatim}
\caption{Example project $controllable$ file for Chinese software projects after discretization. Ranges were assigned randomly for this project. A ``?'' represents a controllable feature. If an attribute range isn't specified in the project, it is ignored.}
\label{fig:china-p2}
\end{figure}

\section{Experiment: {\W}2 vs {\W}}\label{sect:runtimes}

Upon initial experimentation with {\W}, we often followed standard CBR methodology. For example, when deciding how to perform relevancy filtering, we chose the standard CBR practice of taking the euclidean distance from a defined project in n-dimensional space with n project features~\cite{shepperd97}. While this method performed well ~\cite{brady10b}, the $O(n^2)$ runtime requirement prevented us from practically running {\W} on very large datasets.

To resolve this, a simpler method for relevancy was devised. Instead of measuring relevancy based on the distance from a case to the project query's hypervolume, we decided to simply test for inclusion within this volume. The {\em overlap} of a case is simply the number of attributes that fall within the project query's ranges. Because our attributes must be discretized and often rely on qualitative metrics, large overlaps between a query and possible cases are common. The performance of this new method is shown in ~\fig{bestoverlap-vs-knn}. 

{\W}2 is not a slow algorithm. Nothing in any of these steps takes more than  log-linear time, and even that is only to
sort $K_1$ items (which is a very small list).
Even when implemented in an interpreted language  (GAWK), {\W}2 runs in less than half
a second for up to 500 cases (on a 3GHz dual core Macintosh, OS/X 10.6, 4GB of ram).

\begin{figure}[!t]
\begin{center}
\begin{tabular}{|r|r|r|r|r|}\cline{3-4}
 \multicolumn{2}{c}{~}& \multicolumn{2}{|c|}{Execution Time} & \multicolumn{1}{c}{~}\\\hline
dataset     & Cases & W     & W2     & W2 speedup\\\hline
telecom1    & 18        & 0.07s  & 0.04s & 1.6x\\
coc81       & 63        & 0.43s  & 0.08s & 5.3x\\
nasa93      & 93        & 0.69s  & 0.10s & 6.6x\\
china	    & 499       & 7.37s  & 0.42s & 10.8x\\\hline
\end{tabular}
\end{center}
\caption{Average execution times for the W and W2 algorithms. By removing the $O(n^2)$ kth nearest neighbor calculation from W we drastically improve performance, especially on larger datasets such as China (499 cases).}
\label{fig:w2-runtimes}
\end{figure}

\input{figs/knn-vs-bestoverlap}

In all but one case, {\W}2 performs better. However, even when {\W}2 performs slightly worse, it still performs better than KNN in spread reduction. For the Miyazaki dataset, there exists a statistically significant difference (Mann-Whitney, 95\% confidence level).

\section{Experiment: {\W}2's Performance Across Multiple Datasets}\label{sect:multi-comp}

To demonstrate the effectiveness of {\W}2 in any data environment, we offer median reductions for effort reduction for five arbitrary datasets from the PROMISE data repository. The model-agnostic simplicity of {\W}2 made implementing these tests easy as one need only describe a query space and a target utility measure. In the case of these five datasets, software effort was the common target for reduction. 

Given that we did not have access to case studies as we did with NASA93 and COC81 (ground, flight, osp, and osp2) for these datasets, synthetic queries were developed. Three queries were generated for each of the five datasets. The first contained the entire space of possible project attribute values (All), representing complete freedom to recommend any change within the space. The other two queries were generated by randomly choosing 50\% of each attribute values from either the lower, middle, or upper ranges for each project attribute (Proj1, Proj2). These queries represent more common restrictions on possible changes for a given software project.

\begin{figure}[!h]
\begin{center}
\begin{tabular}{|c|c|r|r|}\cline{3-4}
 \multicolumn{2}{c}{~}& \multicolumn{2}{|c|}{Improvement}\\\hline
dataset&query $q$&median&spread\\\hline
Telecom  & Proj1 & 96\% & 23\%\\
Telecom  & Proj2 & 91\% & 41\%\\
Telecom  & All   & 86\% & 28\%\\
Miyazaki & All   & 78\% & 33\%\\
Miyazaki & Proj2 & 69\% & 21\%\\
Miyazaki & Proj1 & 53\% & 67\%\\
Finnish  & All   & 22\% & 31\%\\
Finnish  & Proj2 & 11\% & 27\%\\
Finnish  & Proj1 & 4\% & 25\%\\
China    &  All  & 20\% & 55\%\\
China    & Proj2 & 14\% & 43\%\\
China    & Proj1 & \cellcolor[rgb]{0.8,0.8,0.8}0\% & 13\%\\
Kemerer  & Proj1 & 21\% & 61\%\\
Kemerer  & Proj2 & \cellcolor[rgb]{0.8,0.8,0.8}0\% & 49\%\\
Kemerer  & All   & \cellcolor[rgb]{0.8,0.8,0.8}-4\% & 53\%
\\\hline
\multicolumn{2}{r|}{median} & 21\% & 33\% \\\cline{3-4}
\end{tabular}
\end{center}
\caption{Effort estimation improvements ($100*\frac{initial-final}{intial}$)
for five unique datasets. Sorted by median improvement. Gray cells represent no improvement in effort estimates.}
\label{fig:other-reduce-values}
\end{figure}

\begin{figure}[!t]
\center
\includegraphics[width=5.5in]{figs/noncocomo-reduce.pdf}
\caption{Effort results for five non-COCOMO datasets.
}\label{fig:other-reduce}
\end{figure}

\begin{figure}[!b]
\center
\includegraphics[width=5.5in]{figs/changes.pdf}
\caption{Range of changes in median and spread generated by applying
the recommendations of {\W}2. The median observed
changes were (20.5, 20.5)\% for (medians, spreads), respectively. 
}\label{fig:changes}
\end{figure}


Effort reductions can be seen in ~\fig{other-reduce-values} and ~\ref{fig:other-reduce}. The chart in ~\fig{other-reduce-values} shows strong improvement in median effort for the Telecom and Miyazaki datasets, with strong performance in spread reduction across all datasets. While the Finnish, China, and Kemerer datasets show only marginal or no improvement in median effort, the certainty of their estimations is improved via a reduction in spread.

The other
two queries were generated by randomly choosing 50\% of each attribute
values from either the lower, middle, or upper ranges for each
project descriptor.
For these experiments, the $values$ function was just ``reduce effort''
(the next experiment will explore other results on COCOMO-related data, that tries to reduce effort {\em and} defects {\em and} calendar months).

There are three noteworthy aspects of the \fig{other-reduce} results:
\bi
\item All the points in that figure are positive; i.e.
improvements were seen in all cases.
\item
The dotted lines show the 50\% percentile range of the results: 
half that results had at least
56\% and 73\% improvement in median and spread.
\item
There is no evidence that {\W}2 has problems with
smaller data sets. The two smallest examples processed by {\W}2 are Kemerer and Telecom
containing 15 and 18 examples each. The minimum improvements seen, even for these small data sets,
are 55\% (in both median and spread). 
\ei
The expected value  of the results in these examples is very high; e.g. a 56\% median improvement
in effort. 
The reason for these large improvements is that, in these examples, we focuses {\em only } on effort.
Clearly, there are many ways to cut corners in a project and some of those can have disastrous
results (e.g. allocate no effort to testing will reduce the cost, but that is clearly not a recommended
management action for  a software project). 
Later in this paper are examples where 
{\W}2 is chasing improvements to effort {\em and} defects {\em and} total calendar time to
develop the software. Optimizing for $N=3$ goals is a harder task than just the $N=1$ goal of \fig{other-reduce}.
Hence, those  
the improvements seen in those examples will be more modest (around 20\%).

\section{Experiment: Intra- and Inter-Project Stability}\label{sect:stable}
One of the premises of case-based methods like {\W}2 is that local reasoning
in some specific context is best that fitting one model over an entire space.
This is required if the ``best'' solutions in a one context do not hold in others.

To test this  premise, we generated a report of what treatments were found under different
conditions.
\fig{stability} shows the results of {\W}2's $Step7$ (prune all treatments
that do appear in less than 50\% of 20 repeated trials).
\begin{figure}[!t]
\begin{center}
{\footnotesize
\begin{tabular}{|r|@{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}c@{\hspace{0.75mm}} @{\hspace{0.75mm}}|c@{\hspace{0.75mm}}|c|}\hline
\multicolumn{12}{|c|}{Stability Comparison for NASA93ii FLIGHT}\\\hline
\emph{\begin{sideways}goal\end{sideways}}&
\emph{\begin{sideways}acap=3\end{sideways}}&
\emph{\begin{sideways}apex=3\end{sideways}}&
\emph{\begin{sideways}apex=5\end{sideways}}&
\emph{\begin{sideways}ltex=4\end{sideways}}&
\emph{\begin{sideways}pmat=3\end{sideways}}&
\emph{\begin{sideways}pmat=4\end{sideways}}&
\emph{\begin{sideways}rely=3\end{sideways}}&
\emph{\begin{sideways}sced=2\end{sideways}}&
\emph{\begin{sideways}stor=3\end{sideways}}&
\emph{\begin{sideways}time=3\end{sideways}}&
\emph{\begin{sideways}changes\end{sideways}}\\\hline
defects&  &  &  &  & 90 &  &  &  & 65 & 75 & 3\\
effort &  &  &  &  & 70 &  &  &  & & 90 & 2\\
months &  &  &  &  & 70 &  &  &  & 60 & 85 & 3\\
all &  &  &  &  & 75 &  &  &  & 70 & 85 & 3\\\hline
\multicolumn{12}{|c|}{Stability Comparison for NASA93ii GROUND}\\\hline
\emph{\begin{sideways}goal\end{sideways}}&
\emph{\begin{sideways}acap=3\end{sideways}}&
\emph{\begin{sideways}apex=3\end{sideways}}&
\emph{\begin{sideways}apex=5\end{sideways}}&
\emph{\begin{sideways}ltex=4\end{sideways}}&
\emph{\begin{sideways}pmat=3\end{sideways}}&
\emph{\begin{sideways}pmat=4\end{sideways}}&
\emph{\begin{sideways}rely=3\end{sideways}}&
\emph{\begin{sideways}sced=2\end{sideways}}&
\emph{\begin{sideways}stor=3\end{sideways}}&
\emph{\begin{sideways}time=3\end{sideways}}&
\emph{\begin{sideways}changes\end{sideways}}\\\hline
defects &  &  &  &  & 80 &  &  &  & 50 &  & 2\\
effort &  &  &  &  & 60 &  &  & &  & 75 & 2\\
months &  &  &  &  & 75 &  & 50 &  &  & 75 & 3\\
all &  &  &  &  & 85 &  &  &  & & 80 & 2\\\hline
\multicolumn{12}{|c|}{Stability Comparison for NASA93ii OSP}\\\hline
\emph{\begin{sideways}goal\end{sideways}}&
\emph{\begin{sideways}acap=3\end{sideways}}&
\emph{\begin{sideways}apex=3\end{sideways}}&
\emph{\begin{sideways}apex=5\end{sideways}}&
\emph{\begin{sideways}ltex=4\end{sideways}}&
\emph{\begin{sideways}pmat=3\end{sideways}}&
\emph{\begin{sideways}pmat=4\end{sideways}}&
\emph{\begin{sideways}rely=3\end{sideways}}&
\emph{\begin{sideways}sced=2\end{sideways}}&
\emph{\begin{sideways}stor=3\end{sideways}}&
\emph{\begin{sideways}time=3\end{sideways}}&
\emph{\begin{sideways}changes\end{sideways}}\\\hline
defects & 95 &  & 70 &  &  &  &  & 50 &  &  & 3\\
effort & 80 & 70 &  &  &  &  &  & 70 &  &  & 3\\
months &  &  &  &  &  & 55 &  & 80 &  &  & 2\\
all & 50 &  &  &  &  &  &  & 85 & 60 &  & 3\\\hline
\multicolumn{12}{|c|}{Stability Comparison for NASA93ii OSP2}\\\hline
\emph{\begin{sideways}goal\end{sideways}}&
\emph{\begin{sideways}acap=3\end{sideways}}&
\emph{\begin{sideways}apex=3\end{sideways}}&
\emph{\begin{sideways}apex=5\end{sideways}}&
\emph{\begin{sideways}ltex=4\end{sideways}}&
\emph{\begin{sideways}pmat=3\end{sideways}}&
\emph{\begin{sideways}pmat=4\end{sideways}}&
\emph{\begin{sideways}rely=3\end{sideways}}&
\emph{\begin{sideways}sced=2\end{sideways}}&
\emph{\begin{sideways}stor=3\end{sideways}}&
\emph{\begin{sideways}time=3\end{sideways}}&
\emph{\begin{sideways}changes\end{sideways}}\\\hline
defects &  &  &   & 60 &  & 75 &  & 95 &  &  & 3\\
effort &  &  &  & 75 &  & 55 & & 80 &  &  & 3\\
months &  &  &  & 70 &  & 80 &  & 90 &  &  & 3\\
all &  &  &  & 80 &  & 90 &  & 100 & &  & 3\\\hline
\end{tabular}}
\end{center}
\caption{Recommendation frequency across 20 runs of {\W}2 for reducing individual goals 
($defects$, $effort$, or $months$) as well as all goals at once ($all$).}
\label{fig:stability}
\end{figure}
The left-hand column of \fig{stability} shows the four $values$ function used
in that study:
\be
\item {\em Defects} aims at reducing just defects;
\item {\em Effort} aims at  reducing just effort;
\item {\em Months} aims at reducing
a project's total calendar time.
\item {\em All} refers to \eq{edt}; i.e. try to decrease effort {\em and} development time
{\em and} number of defects;
\ee
The last of these is a multi-objective function while the rest strive to optimize
one objective without concern to the others.
\fig{stability} shows that,
in any row, the conclusions reached by {\W}2 are stable (i.e. appear at high frequency,
across 20 random selections of $train:test$). 
That is, {\W}2's results exhibit
$intra-$project conclusion stability (when the
$context$ and $values$ function are held constant).
For project managers, this is good news since it shows that their data contains
clear signals on how to best change their particular project in order to achieve particular
goals.

However, \fig{stability}  also shows that  
if the $context$ is changed (from generalized FLIGHT systems to a specific
flight system like OSP), then the recommended changes are very different.
Similarly, the OSP results show that
altering the $values$ function also dramatically changes recommendations.

Menzies \& Shull~\cite{me10b} report that many SE papers conclude that
what is ``best'' for one project may not
be ``best'' for another.
For example, Zimmermann studied 629 pairs of software development 
projects~\cite{zimmermann09}. In only 4\% of those hundreds of pairs was a 
defect prediction model learned from one project useful on 
another. 
When such {\em inter-}project conclusion instability exists,
 then
tools like {\W}2 are essential since
  it is best  to learn changes that are tuned to the specifics of particular projects (like OSP \&
OSP2) rather than on generalized descriptions of software (like FLIGHT \& GROUND).

\section{Experiment: Comparing Drastic Changes to {\W}2}\label{sect:drastic-comp}

Prior work considered explored impact of so-called {\em drastic} changes to software projects~\cite{me09a}. A drastic change occurs when the recommendation falls outside the defined projects ranges of a software project. In other words, when the recommended course of action is dramatic. For example, with the OSP NASA case study (~\fig{cases}), attempting to improve programmer language and tool experience (ltex) to 5 (very high, 6+ years experience), would be a drastic change as the maximum defined value for ltex is 4 (high). 

To test this, {\W}2's recommendations for the four NASA case studies (~\fig{cases}) in the NASA93 dataset were overridden with the drastic changes from ~\fig{drastic-overview}. For 3 distinct drastic changes, {\W}2 attempted to apply the drastic changes until no improvement was measured, then reported the median effort, defects, and months for that change. Note that not all changes from ~\cite{me09a} were applicable, due to a lack of extreme cases in the NASA93 dataset.

The results for these changes are reported in Figures ~\ref{fig:drastic-flight},~\ref{fig:drastic-ground},~\ref{fig:drastic-osp},~\ref{fig:drastic-osp2}. Of the 12 comparisons, in only one case does {\W}2 perform better statistically and significantly better than the three drastic changes. However, in terms of median reductions, {\W}2 always performs in the top 50\% of of cases. Most importantly, even when compared to extreme project recommendations, {\W}2 is not constrained by the limited project attribute ranges allowed for its recommendations.

\begin{figure}[!t]
\begin{center}
\begin{tabular}{c l l}
  & \textbf{Drastic Change}   & \textbf{Attribute Effects}\\\hline
1 & Improve Process Maturity  & pmat = 5\\\hline
2 & Improve Tools\&Techniques & time = 3; stor = 3;\\
  &                           & pvol = 2; tool = 5; site = 6\\\hline
3 & Reduce Functionality      & data = 2;\\\hline
\end{tabular}
\end{center}
\caption{Examples of drastic changes to software projects.}
\label{fig:drastic-overview}
\end{figure}

\input{figs/drastic.tex}